\begin{table}%tB.1
%\centering
\par
\caption {\label{tabb1}Analyzed lines in the stellar spectra, measured equivalent widths
in m\AA, and relative abundances.\ion{He}{i} is not included.} 
\begin{tabular}{llrrrccccrr}
\hline\hline
\multicolumn{5}{c}{}&\multicolumn{2}{c}{HR~6000[13450, 4.3, AT12]}&
\multicolumn{2}{c}{46~Aql[12560, 3.8, AT12]}
\\
\hline\noalign{\smallskip}
\multicolumn{1}{c}{Species}&
\multicolumn{1}{c}{$\lambda$($\AA$)}&
\multicolumn{1}{c}{$\log~gf$}&
\multicolumn{1}{c}{Ref.$^{a}$}&
\multicolumn{1}{c}{$\chi_{\rm low}$}&
\multicolumn{1}{c}{$W$(m$\AA$)}&
\multicolumn{1}{c}{$\log(N_{\rm elem}/N_{\rm tot}$)}&
\multicolumn{1}{c}{$W$(m$\AA$)}&
\multicolumn{1}{c}{$\log(N_{\rm elem}/N_{\rm tot}$)}
\\
\hline\noalign{\smallskip}
\ion{Be}{ii} & 3130.420 & $-$0.170 & NIST3 &    0.00  & 26.7   & $-$9.78 & 31.0  & $-$9.91\\  
\ion{C}{ii}  & 4267.001 & $+$0.562 &NIST3&145549.270  &profile & $-$5.50  &profile& $-$4.75 \\
\ion{C}{ii}  & 4267.261 & $+$0.716 &NIST3&145550.700  &profile & $-$5.50  &profile& $-$4.75 \\
\ion{N}{i}   &8680.282  & $+$0.347 &NIST3& 83364.620  & not obs& $\le$$-$5.80 &not obs& $\le$$-$5.50\\
\ion{N}{i}   &8683.403  & $+$0.087 &NIST3& 88317.830  & not obs& $\le$$-$5.80 &not obs& $\le$$-$5.50\\
\ion{O}{i}   &3947.295  & $-$2.095 &NIST3&73768.200   &profile & $-$3.71 & profile & $-$3.71\\
\ion{O}{i}   &3947.481  & $-$2.244 &NIST3&73768.200   &profile & $-$3.71 & profile & $-$3.71\\
\ion{O}{i}   &3947.586  & $-$2.467 &NIST3&73768.200   &profile & $-$3.71 & profile & $-$3.71\\
\ion{O}{i}   &4368.193  & $-$2.665 &NIST3&76794.978   &profile & $-$3.71 & profile & $-$3.51\\
\ion{O}{i}   &4368.242  & $-$1.964 &NIST3&76794.978   &profile & $-$3.71 & profile & $-$3.51\\
\ion{O}{i}   &4368.258  & $-$2.186 &NIST3&76794.978   &profile & $-$3.71 & profile & $-$3.51\\
\ion{O}{i}   &5329.096  & $-$1.938 &NIST3&86625.757   &profile & $-$3.71 & profile & $-$3.46\\
\ion{O}{i}   &5329.099  & $-$1.586 &NIST3&86625.757   &profile & $-$3.71 & profile & $-$3.46\\
\ion{O}{i}   &5329.107  & $-$1.695 &NIST3&86625.757   &profile & $-$3.71 & profile & $-$3.46\\
\ion{O}{i}   &6155.961  & $-$1.363 &NIST3&86625.757   &profile & $-$3.64 & profile & $-$3.46\\
\ion{O}{i}   &6155.971  & $-$1.011 &NIST3&86625.757   &profile & $-$3.64 & profile & $-$3.46\\
\ion{O}{i}   &6155.989  & $-$1.120 &NIST3&86625.757   &profile & $-$3.64 & profile & $-$3.46\\
\ion{O}{i}   &6156.737  & $-$1.487 &NIST3&86627.778   &profile & $-$3.64 & profile & $-$3.43\\
\ion{O}{i}   &6156.755  & $-$0.898 &NIST3&86627.778   &profile & $-$3.64 & profile & $-$3.43\\
\ion{O}{i}   &6156.778  & $-$0.694 &NIST3&86627.778   &profile & $-$3.64 & profile & $-$3.43\\
\ion{O}{i}   &6455.977  & $-$0.920 &NIST3&86631.454  &profile  & $-$3.59 & profile & $-$3.43\\
\ion{O}{i}   &7002.196  & $-$1.489 &NIST3&88631.146 &profile & $-$3.71 &  H$_{2}$O &$-$\\
\ion{O}{i}   &7002.230  & $-$0.741 &NIST3&88631.146 &profile & $-$3.71 & H$_{2}$O  & $-$\\
\ion{O}{i}   &7002.250  & $-$1.364 &NIST3&88631.303 &profile & $-$3.71 & H$_{2}$O  & $-$\\
\ion{Ne}{i}  &7032.413  & $-$0.249 &NIST3&134041.840& noise? & $-$4.86 & noise ? &$-$4.51 \\
\ion{Na}{i}  &5688.205  & $-$0.452 &NIST3&16973.368 &not obs & $\le$$-$5.71& profile & $-$5.65\\
\ion{Na}{i}  &5889.950  & $+$0.108 &NIST3& 0.00     &inters. & $-$      & profile & $-$5.67\\
\ion{Na}{i}  &5895.924  & $-$0.194 &NIST3& 0.00     &inters. & $-$      & profile & $-$5.72\\
\ion{Mg}{ii} &4481.126  & $+$0.749 &NIST3&71490.190 &profile &$-$5.66    & profile & $-$5.45\\
\ion{Mg}{ii} &4481.150  & $-$0.553 &NIST3&71490.190 &profile &$-$5.66    & profile & $-$5.45\\
\ion{Mg}{ii} &4481.325  & $+$0.594 &NIST3&71491.063 &profile &$-$5.66    & profile & $-$5.45\\
\ion{Al}{i}  &3944.006  & $-$0.638 &NIST3&    0.000 &not obs &$\le$$-$7.30&not obs & $\le$$-$6.65\\
\ion{Al}{i}  &3961.520  & $-$0.336 &NIST3&  112.061 &not obs &$\le$$-$7.30&profile &   $-$6.65 \\
\ion{Al}{ii} &7042.083  & $+$0.340 &NIST3  &91274.500 &noise ? &$\le$$-$7.30&noise ? &$\le$$-$7.40\\
\ion{Al}{ii} &7056.712  & $+$0.110 &NIST3  &91274.500 &noise ? &$\le$$-$7.30&noise ? &$\le$$-$7.00\\
\ion{Si}{ii} &3853.665  & $-$1.341 &NIST3  &55309.350&not obs &$-$ & 31.8& $-$5.62\\
\ion{Si}{ii} &3856.018  & $-$0.406 &NIST3  &55325.180&not obs &$-$ & 71.8 & $-$5.68\\
\ion{Si}{ii} &3862.595  & $-$0.757 &NIST3  &55309.350&profile &$-$7.35 & 62.6 & $-$5.53\\
\ion{P}{ii}& 4044.576 & $+$0.481 &K, MRB & 107360.250 & 45.3 & $-$4.42 & 19.9 & $-$5.16\\
\ion{P}{ii} & 4062.149 &   $-$0.585 &K, MRB & 103339.140 & 8.6 & $-$4.79 & 2.4 & $-$5.45\\
\ion{P}{ii}  & 4127.559 &   $-$0.110 &K, KP & 103667.860 & 25.2 & $-$4.54 &9.3 & $-$5.21\\
\ion{P}{ii}  & 4160.623 &   $-$0.410 &K, KP & 103339.140 & 9.4 & $-$4.90 & 7.6 & $-$5.03\\
\ion{P}{ii}  & 4244.622 &   $-$0.308 &K, MRB & 107360.250 &9.6 & $-$4.78 &2.6 & $-$5.46\\
\ion{P}{ii}  & 4288.606 &   $-$0.630 &K, MRB & 101635.690 & 7.0 & $-$4.88&blend& $-$\\
\ion{P}{ii}  & 4420.712 &   $-$0.329 &NIST3 & 88893.220  & 40.1 & $-$4.51&19.4& $-$5.18\\
\ion{P}{ii}  & 4452.472 &   $-$0.194 &K, MRB & 105302.170  & 19.9 & $-$4.48 &6.9 & $-$5.13\\
\ion{P}{ii}  & 4463.027 &   $+$0.026 &K, MRB & 105549.670  & 25.6 & $-$4.52 &9.1 & $-$5.19\\
\ion{P}{ii}  & 4466.140 &   $-$0.560 &NIST3 & 105549.670  & 9.8 & $-$4.57 & 2.2 & $-$5.32\\
\ion{P}{ii}  & 4468.000 &   $-$0.208 &K, MRB & 105244.060  & 24.0 & $-$4.35 &blend&  $-$\\
\ion{P}{ii}  & 4475.270 &   $+$0.440 &NIST3 & 105549.670   & 36.8 & $-$4.63 & 16.2 & $-$5.25 \\
\ion{P}{ii}  & 4483.693 &   $-$0.431 &NIST3 & 105302.370   & 10.0 & $-$4.69 & 2.3 & $-$5.43\\
\ion{P}{ii}  & 4499.230 &   $+$0.470 &NIST3 & 107922.930   &38.0 & $-$4.44 & blend &$-$\\
\ion{P}{ii}  & 5296.077 &   $-$0.160 &NIST3 & 87124.600    &56.3 & $-$4.15 & 33.9 & $-$4.74\\
\ion{P}{ii}  & 5316.055 &   $-$0.294 &NIST3 & 86743.960    & 47.9 & $-$4.31 & 28.3 & $-$4.82\\
\ion{P}{ii}  & 5344.729 &   $-$0.390 &NIST3 & 86597.550    & 45.6 & $-$4.28 &25.7 & $-$4.81\\
\ion{P}{ii}  & 5386.895 &   $-$0.520 &NIST3 & 86743.960    & 51.8 & $-$3.98 &25.6 & $-$4.67\\
\ion{P}{ii}  & 5425.880 &   $+$0.180 &NIST3 & 87124.600    & 71.8 & $-$4.22 & 47.4 & $-$4.71\\
\hline
\ion{P}{ii}  & 6024.178 &   $+$0.137 &NIST3 & 86743.960    & 69.8 & $-$4.04 & 45.7 & $-$4.57\\
\ion{P}{ii}  & 6034.039 &   $-$0.220 &NIST3 & 86597.550    & 52.6 & $-$4.10 & 29.4 & $-$4.70\\
\ion{P}{ii}  & 6043.084 &   $+$0.416 &NIST3 & 87124.600    & 81.1 & $-$4.05 & 55.5 & $-$4.57\\
\ion{P}{iii}& 4059.312 &   $-$0.051 &NIST3 &116885.870     &11.1 & $-$4.97  & 2.5 & $-$5.87\\
\ion{P}{iii} & 4080.089 &   $-$0.306 &NIST3 &116874.560     & 5.5 & $-$5.21 & not ob & $-$\\
\ion{P}{iii} & 4222.198 &   $+$0.210 &NIST3 &117835.950     &19.5 & $-$5.45 & not obs &$-$\\
\ion{P}{iii} & 4246.720 &   $-$0.120 &NIST3 &117835.950     & 10.6 & $-$4.78 & not obs & $-$\\
\ion{S}{ii}  & 4162.665 &   $+$0.777 &NIST3 &128599.160      & 2.0 & $-$6.26 & 5.3 & $-$5.74\\
\ion{Cl}{i}  & 4794.556 &   $+$0.400 &NIST3 &107879.660   & not obs &$\le$$-$7.24& not obs& $\le$$-$7.04\\
\ion{Ca}{i}  & 4226.728 &   $+$0.244 &NIST3 &     0.000    &profile &$>$$-$5.68 & profile & $>$$-$5.98 \\       
\ion{Ca}{ii} & 3933.663 &   $+$0.135 &NIST3 &     0.000    & profile &$-$5.68 & profile &$-$5.98\\       
\ion{Ca}{ii} & 3968.469 &   $-$0.180 &NIST3 &     0.000    & profile &$-$5.68 & profile & $-$5.98\\       
\ion{Sc}{ii} & 4246.822 &   $+$0.242 &NIST3 &   2540.950    & blend & $-$ & blend & $-$  \\       
\ion{Sc}{ii} & 4314.083 &   $-$0.100 &NIST3 &   4987.790    & not obs &$\le$$-$9.50 & not obs &$\le$$-$9.50\\       
\ion{Ti}{ii} & 4053.821 &   $-$1.130 &PTP   &  15265.620 & 6.4 & $-$6.40  & 11.1 & $-$6.46\\       
\ion{Ti}{ii} & 4163.644 &   $-$0.130 &PTP   &  20891.660 & 19.3 & $-$6.51 & 30.4 & $-$6.50 \\       
\ion{Ti}{ii} & 4287.873 &   $-$1.790 &PTP   &  8710.440  & 3.5 & $-$6.38  &  6.1 & $-$6.49\\       
\ion{Ti}{ii} & 4290.215 &   $-$0.850 &PTP   &  9395.710  & 15.6 &$-$6.55  & 28.3 & $-$6.52\\       
\ion{Ti}{ii} & 4294.094 &   $-$0.930 &PTP   &  9744.250  & 16.5 & $-$6.47 & 28.3 & $-$6.48\\       
\ion{Ti}{ii} & 4300.042 &   $-$0.440 &PTP   &  9518.060  & 30.2 & $-$6.52 & 45.2 & $-$6.49\\       
\ion{Ti}{ii} & 4301.922 &   $-$1.150 &PTP   &  9363.620  & 8.6 &  $-$6.56 & 17.1 & $-$6.55\\       
\ion{Ti}{ii} & 4367.652 &   $-$0.860 &PTP   & 20891.660  & 5.7 &  $-$6.42 &  9.9 & $-$6.46\\       
\ion{Ti}{ii} & 4395.031 &   $-$0.540 &PTP   &  8744.250  &29.2 &  $-$6.49 & 45.2 & $-$6.43\\       
\ion{Ti}{ii} & 4399.765 &   $-$1.190 &PTP   &  9975.920  &3.3 &   $-$6.94 & 17.2 & $-$6.48\\       
\ion{Ti}{ii} & 4411.072 &   $-$0.670 &PTP   & 24961.030  &5.0 &   $-$6.45 & 10.1 & $-$6.40\\       
\ion{Ti}{ii} & 4417.714 &   $-$1.190 &PTP   &  9395.710  &8.8 &   $-$6.51 & 16.4 & $-$6.54\\       
\ion{Ti}{ii} & 4418.331 &   $-$1.970 &PTP   &  9975.920  &2.4 &   $-$6.31 &  3.5 & $-$6.50\\       
\ion{Ti}{ii} & 4443.810 &   $-$0.720 &PTP   &  8710.440  &22.3 &  $-$6.50 & 37.1 & $-$6.47\\       
\ion{Ti}{ii} & 4450.482 &   $-$1.520 &PTP   &  8744.250  &5.5 &   $-$6.45 & 11.5 & $-$6.44\\       
\ion{Ti}{ii} & 4464.448 &   $-$1.810 &PTP   &  9363.620  &3.1 &   $-$6.40 &  6.8 & $-$6.38\\       
\ion{Ti}{ii} & 4468.492 &   $-$0.620 &NIST3 &  9118.260  &25.6 &  $-$6.50 & 39.6 & $-$6.50\\       
\ion{Ti}{ii} & 4488.325 &   $-$0.510 &PTP   & 25192.710  &7.0 &   $-$6.44 & 12.4 & $-$6.44\\       
\ion{Ti}{ii} & 4501.270 &   $-$0.770 &PTP   &  8997.710  &23.0 &  $-$6.41 & 37.6 & $-$6.39\\       
\ion{Ti}{ii} & 4805.085 &   $-$1.120 &NIST3 & 16625.110  &7.2 &   $-$6.28 & 14.5 & $-$6.27\\       
\ion{Ti}{ii} & 4911.195 &   $-$0.610 &PTP   & 25192.790  &5.4 &   $-$6.45 & 10.7 & $-$6.42\\       
\ion{V}{ii}  & 3093.105 &   $+$0.530 &K, NBS   &  3162.800  &blend &$\le$$-$9.14 & noise ?& $\le$$-$8.94   \\       
\ion{V}{ii}  & 3102.289 &   $+$0.410 &K, NBS   &  2968.220  &not obs & $\le$$-$9.14  &noise &$\le$$-$8.94 \\       
\ion{V}{ii}  & 3110.695 &   $+$0.300 &K, NBS   &  2808.720  &blend &$\le$$-$9.14  &blend  & $\le$$-$8.94\\       
\ion{V}{ii}  & 3113.376 &   $+$0.180 &K, NBS   &  2687.010  &not obs & $\le$$-$9.14 &not obs &$\le$$-$8.94 \\       
\ion{V}{ii}  & 3125.276 &   $+$0.070 &K, NBS   &  2604.820  &blend &   $\le$$-$9.14 &blend & $-$8.94 \\       
\ion{Cr}{ii} & 4812.337 &   $-$1.800 &NIST3   &  31168.580  &3.7 &$-$6.21& not obs & $-$ \\       
\ion{Cr}{ii} & 4824.127 &   $-$0.970 &SL      &  31219.350  &26.4 &$-$5.98 & blend & $-$\\       
\ion{Cr}{ii} & 4836.229 &   $-$2.000 &SL      &  31117.390  &3.9 &$-$5.99  &not obs  & $-$\\       
\ion{Cr}{ii} & 5237.329 &   $-$1.160 &NIST3   &  32854.310  &12.5 &$-$6.16  & 1.5 & $-$7.48 \\       
\ion{Cr}{ii} & 5246.768 &   $-$2.460 &NIST3   &  29951.880  &1.2 &$-$6.14  &not obs  & $-$\\       
\ion{Mn}{ii} & 3441.988 &   $-$0.270 &NIST3   &  14325.860  &85.3 &$-$4.92 ($-$5.00)$^{b}$  &74.5  & $-$5.48 ($-$5.35)$^{b}$\\       
\ion{Mn}{ii} & 3460.316 &   $-$0.540 &NIST3   &  14593.820  &74.5 &$-$4.91 ($-$5.00)  &60.2  & $-$5.65 ($-$5.65)\\       
\ion{Mn}{ii} & 3482.905 &   $-$0.740 &NIST3   &  14781.190  &65.3 &$-$4.98 ($-$5.00)  &53.0  & $-$5.70 ($-$5.65)\\       
\ion{Mn}{ii} & 3488.677 &   $-$0.860 &NIST3   &  14901.180  &65.7 &$-$4.85 ($-$4.85)  &52.5  & $-$5.60 ($-$5.60)\\       
\ion{Mn}{ii} & 3495.833 &   $-$1.220 &NIST3   &  14959.840  &55.3 &$-$4.85 ($-$4.85)  &42.2  & $-$5.65 ($-$5.65) \\       
\ion{Mn}{ii} & 3496.809 &   $-$1.690 &NIST3   &  14781.190  &38.1 &$-$5.09 ($-$5.00)  &27.0  & $-$5.87 ($-$5.90)\\       
\ion{Mn}{ii} & 3497.526 &   $-$1.330 &NIST3   &  14901.180  &47.7 &$-$5.05 ($-$4.90)  &35.5  & $-$5.84 ($-$5.85)\\       
\ion{Mn}{ii} & 3917.318 &   $-$1.169 &K03Mn   &  55759.270  &4.9 &$-$5.61 ($-$5.65)   &2.8  & $-$6.09  ($-$6.17)\\       
\ion{Mn}{ii} & 4363.258 &   $-$1.929 &K03Mn   &  44899.820  &4.2 &$-$5.49 ($-$5.58)  & 2.3  & $-$6.03 ($-$6.15)\\       
\ion{Mn}{ii} & 4365.219 &   $-$1.339 &K03Mn   &  44899.820  &4.7 &$-$5.59 ($-$5.63)  & 3.1  & $-$6.02 ($-$6.07)\\       
\ion{Mn}{ii} & 4478.635 &   $-$0.942 &K03Mn   &  53597.130  &8.2 &$-$5.68 ($-$5.63)  & 5.0  & $-$6.16 ($-$6.15)\\       
\ion{Fe}{i}  & 3581.193 &   $+$0.406 &FW06 & 6928.27  & 17.3 & $-$3.67 & 19.2& $-$3.86\\
\ion{Fe}{i}   & 3618.768 &   $-$0.003 &FW06 & 7985.78  & 8.0 & $-$3.73  & 10.3& $-$3.88\\
\ion{Fe}{i}   & 4005.242 &   $-$0.610 &FW06 &12560.93  & 6.6 & $-$3.59  &  6.4& $-$3.94\\
\hline
\ion{Fe}{i}   & 4071.738 &   $-$0.022 &FW06 &12698.55  &15.4 &  $-$3.71 & 17.0& $-$3.99\\
\ion{Fe}{i}   & 4202.029 &   $-$0.708 &FW06 &11976.24  & 4.5 &  $-$3.71 &  5.7& $-$3.93\\
\ion{Fe}{i}   & 4219.360 &   $+$0.000 &FW06 &28819.95  & 4.0 &  $-$3.55 &  4.4& $-$3.78\\
\ion{Fe}{i}   & 4235.936 &   $-$0.341 &FW06 &19562.44  & 4.9 &  $-$3.62 &  5.9& $-$3.84\\
\ion{Fe}{i}   & 4271.760 &   $-$0.164 &FW06 &11976.24  &14.7 &  $-$3.66 & 16.4& $-$3.93\\
\ion{Fe}{i}   & 4383.545 &   $+$0.200 &FW06 &11976.24  &25.2 &  $-$3.67 & 27.6& $-$3.95\\
\ion{Fe}{i}   & 4404.750 &   $-$0.142 &FW06 &12560.93  &18.6 &  $-$3.51 & 18.0& $-$3.87\\
\ion{Fe}{i}   & 4415.122 &   $-$0.615 &FW06 &12968.55  & 6.2 &  $-$3.61 &  8.0& $-$3.81\\
\ion{Fe}{i}   & 5324.179 &   $-$0.103 &FW06 &25899.99  & blend&  $-$    &  3.3& $-$3.98\\
\ion{Fe}{i}   & 5364.871 &   $+$0.228 &FW06 &35856.40  &2.4  &  $-$3.61  &  2.2& $-$3.92\\
\ion{Fe}{i}   & 5369.962 &   $+$0.536 &FW06 &35257.32  &5.0  &  $-$3.62  &  4.3& $-$3.95\\
\ion{Fe}{i}   & 5410.910 &   $+$0.398 &FW06 &36079.37  &2.6  &  $-$3.74  &  2.9& $-$3.94\\
\ion{Fe}{i}   & 5424.068 &   $+$0.580 &FW06 &34843.95  &blend& $-$      &  5.0& $-$3.94\\
\ion{Fe}{i}   & 5615.644 &   $+$0.050 &FW06 &26874.55  &blend &$-$      &  3.9& $-$3.99\\ 
\ion{Fe}{ii}  & 3906.035 &   $-$1.700 &FW06 &44929.55  &42.8 & $-$3.82  & 43.7 & $-$4.02\\
\ion{Fe}{ii}   & 3914.503 &   $-$4.370 &FW06 &13473.41  &18.1 &  $-$3.60 & 22.4 & $-$3.80\\
\ion{Fe}{ii}   & 3935.962 &   $-$1.720 &FW06 &44915.05  &43.3 & $-$3.79 & 44.1 & $-$3.99\\
\ion{Fe}{ii}   & 3938.290 &   $-$4.070 &FW06 &13471.41  &37.1 &  $-$3.35 & 31.7 & $-$3.83\\
\ion{Fe}{ii}   & 3945.210 &   $-$4.440 &FW06 &13673.18  &14.7 &  $-$3.65 & 18.1 & $-$3.85\\
\ion{Fe}{ii}   & 4122.668 &   $-$3.300 &FW06 &20830.58 &33.4 &  $-$3.82  & 36.8 & $-$4.03\\
\ion{Fe}{ii}   & 4128.748 &   $-$3.580 &FW06 &20830.58 &23.6 &   $-$3.82 & 27.9 & $-$4.00\\
\ion{Fe}{ii}   & 4178.862 &   $-$2.440 &FW06 &20830.58 &58.3 &  $-$3.93  & 61.7 & $-$4.11\\
\ion{Fe}{ii}   & 4258.154 &   $-$3.480 &FW06 &21812.05 &26.6 &   $-$3.78 & 29.7 &$-$3.99\\
\ion{Fe}{ii}   & 4273.326 &   $-$3.300 &FW06 &21812.05 &32.6 &  $-$3.79  & 39.6 &$-$3.90\\
\ion{Fe}{ii}   & 4296.572 &   $-$2.930 &FW06 &21812.05 &44.5 &  $-$3.82  & 50.1 &$-$3.95\\
\ion{Fe}{ii}   & 4303.176 &   $-$2.610 &FW06 &21812.05 &57.6 &  $-$3.74  & 63.1 &$-$3.86\\
\ion{Fe}{ii}   & 4369.411 &   $-$3.580 &FW06 &22409.85 &21.5 &   $-$3.80 & 25.5 &$-$3.98\\
\ion{Fe}{ii}   & 4413.601 &   $-$4.190 &FW06 &21581.64 &10.4&   $-$3.66 & 13.7 &$-$3.81\\
\ion{Fe}{ii}   & 4416.830 &   $-$2.600 &FW06 &22409.85 &56.7 &  $-$3.75  & 59.3 &$-$3.96\\
\ion{Fe}{ii}   & 4491.405 &   $-$2.640 &FW06 &23031.30 &49.8 &  $-$3.88  & 53.5 &$-$4.06\\
\ion{Fe}{ii}   & 4508.288 &   $-$2.350 &FW06 &23031.30 &66.2 &  $-$3.68  & 70.1 &$-$3.84\\
\ion{Fe}{ii}   & 4515.339 &   $-$2.360 &FW06 &23939.36 &57.8 &  $-$3.86  & 61.3 &$-$4.04\\
\ion{Fe}{ii}   & 4520.224 &   $-$2.620 &FW06 &22637.21 &54.4 &  $-$3.78 &no spectra&$-$ \\
\ion{Fe}{ii}   & 4522.634 &   $-$1.990 &FW06 &22939.36 &68.7 &  $-$3.97   &no spectra&$-$\\
\ion{Fe}{ii}   & 4923.927 &   $-$1.210 &FW06 &23317.63 &100.8 &  $-$3.91  &blend&$-$\\
\ion{Fe}{ii}   & 4993.358 &   $-$3.680 &FW06 &22637.20 &19.6 &   $-$3.74 & 23.2 &$-$3.93\\
\ion{Fe}{ii}   & 5004.188 &   $+$0.482 &K09&82853.66 &      &           & 47.6 &$-$3.88\\
\ion{Fe}{ii}   & 5006.840 &   $-$0.281 &K09&83713.53 &      &           & 20.4 &$-$3.90\\
\ion{Fe}{ii}   & 5007.450 &   $-$0.382 &K09&83726.37 &      &           & 16.4 &$-$3.94\\
\ion{Fe}{ii}   & 5010.060 &   $-$0.808 &K09&83726.37 &      &           & 10.1 &$-$3.80\\
\ion{Fe}{ii}   & 5011.026 &   $-$1.177 &K09&83726.37 &      &           &  3.6 &$-$3.95\\ 
\ion{Fe}{ii}   & 5018.440 &   $-$1.350 &FW06 &23317.63 &111.7 &  $-$3.57 &116.7  &$-$3.77\\
\ion{Fe}{ii}   & 5022.419 &   $-$0.052 &K09&83459.67 &      &           & 23.6 &$-$4.03\\
\ion{Fe}{ii}   & 5026.798 &   $-$0.235 &K09&83136.46 &      &           & 22.0 &$-$3.92\\
\ion{Fe}{ii}   & 5030.631 &   $+$0.431 &FW06 &82978.68 &50.7  &   $-$3.69 & 43.2 &$-$3.95\\
\ion{Fe}{ii}   & 5032.704 &   $+$0.143 &K09&83812.30 &      &           & 30.9 &$-$3.97\\
\ion{Fe}{ii}   & 5035.700 &   $+$0.630 &FW06 &82978.68 &59.7  &   $-$3.64 & 51.6 &$-$3.90\\
\ion{Fe}{ii}   & 5036.713 &   $-$0.527 &K09&83812.30 &      &           & 13.0 &$-$3.94\\
\ion{Fe}{ii}   & 5045.108 &   $-$0.116 &K09&83136.46 &      &           & 26.8 &$-$3.88\\
\ion{Fe}{ii}   & 5060.249 &   $-$0.479 &K09&84266.54 &      &           & 12.7 &$-$3.97\\
\ion{Fe}{ii}   & 5062.927 &   $-$1.166 &K09&83136.46 &      &           &  7.0 &$-$3.68\\
\ion{Fe}{ii}   & 5067.890 &   $-$0.173 &K09&83308.19 &      &           & 22.4 &$-$3.95\\
\ion{Fe}{ii}   & 5070.583 &   $-$0.865 &K09&83308.19 &      &           &  6.6 &$-$3.99\\
\ion{Fe}{ii}   & 5070.895 &   $+$0.262 &K09&83136.46 &      &           & 37.9 &$-$3.92\\
\ion{Fe}{ii}   & 5075.760 &   $+$0.233 &K09&84326.92 &      &           & 33.3 & $-$3.95\\
\ion{Fe}{ii}      & 5076.597 &   $-$0.749 &K09&83713.53 &      &           &  8.8 & $-$3.95\\
\ion{Fe}{ii}      & 5081.898 &   $-$0.689 &K09&83713.53 &      &           &  8.3 & $-$4.02\\
\hline
\ion{Fe}{ii}      & 5086.306 &   $-$0.472 &K09&83990.06 &      &           & 14.7 & $-$3.90\\
\ion{Fe}{ii}      & 5089.214 &   $-$0.014 &K09&83308.19 &      &           & 27.0 & $-$3.96\\
\ion{Fe}{ii}      & 5093.783 &   $-$0.703 &K09&83726.37 &      &           & 12.7 & $-$3.77\\
\ion{Fe}{ii}      & 5117.032 &   $-$0.129 &K09&84131.58 &      &           & 22.0 & $-$3.96\\    
\ion{Fe}{ii}      & 5132.669 &   $-$4.090 &FW06 &22637.20 &17.5 &   $-$3.40 & 14.0 & $-$3.83\\
\ion{Fe}{ii}      & 5140.692 &   $-$0.822 &K09&83990.06 &      &           &  9.9 & $-$3.78\\
\ion{Fe}{ii}      & 5143.875 &   $+$0.054 &K09&84266.54 &      &           & 21.6 & $-$4.14\\
\ion{Fe}{ii}      & 5144.352 &   $+$0.307 &FW06 &84424.37 &40.1 &   $-$3.73 & 32.8 & $-$4.03\\
\ion{Fe}{ii}      & 5169.033 &   $-$0.870 &FW06 &23317.63 &115.1 &  $-$4.01 & 121.0 & $-$4.19\\
\ion{Fe}{ii}      & 5180.312 &   $+$0.002 &K09&83812.30 &       &          & 26.9 & $-$3.94\\
\ion{Fe}{ii}      & 5199.118 &   $+$0.054 &K09&83713.53 &       &          & 30.3 & $-$3.87\\
\ion{Fe}{ii}      & 5200.798 &   $-$0.390 &K09&83812.30 &       &          & 17.4 & $-$3.87\\
\ion{Fe}{ii}      & 5203.634 &   $-$0.088 &K09&83812.30 &       &          & 25.9 & $-$3.88\\
\ion{Fe}{ii}      & 5218.841 &   $-$0.250 &K09&83726.37 &       &          & 23.6 & $-$3.85\\
\ion{Fe}{ii}      & 5219.920 &   $-$0.628 &K09&84870.86 &       &          & 10.3 & $-$3.89\\
\ion{Fe}{ii}      & 5222.350 &   $-$0.332 &K09&84844.82 &       &          & 15.0 & $-$3.96\\
\ion{Fe}{ii}      & 5223.802 &   $-$0.546 &K09&83713.53 &       &          & 14.6 & $-$3.83\\
\ion{Fe}{ii}      & 5224.404 &   $-$0.581 &K09&83990.06 &       &          & 12.5 & $-$3.87\\    
\ion{Fe}{ii}      & 5234.625 &   $-$2.210 &FW06&25981.63 &69.6 &  $-$3.54  & 71.0 & $-$3.76\\
\ion{Fe}{ii}      & 5237.949 &   $+$0.103 &K09&84266.54 &       &          & 29.8 & $-$3.90\\
\ion{Fe}{ii}      & 5245.455 &   $-$0.502 &K09&84326.92 &       &          & 13.6 & $-$3.88\\
\ion{Fe}{ii}      & 5247.956 &   $+$0.550 &FW06 &84938.18 &46.2 &  $-$3.74 & 39.7 & $-$4.01\\
\ion{Fe}{ii}      &  5257.119 &  $+$0.115 &K07 &84685.20 &      &          & 26.7 & $-$4.00\\
\ion{Fe}{ii}      &  5265.985 &  $-$0.936 &K09 &84035.12 &      &          &  9.3 & $-$3.68\\
\ion{Fe}{ii}      &  5270.029 &  $-$0.097 &K09 &84710.70 &      &          & 18.0 & $-$4.08\\
\ion{Fe}{ii}      & 5272.397 &   $-$2.010 &FW06 &48039.09 &35.0 &   $-$3.51  & 35.4 & $-$3.71\\
\ion{Fe}{ii}      & 5276.002 &   $-$1.900 &FW06 &25805.33 &67.0 &  $-$3.93   & 71.7 & $-$4.07\\
\ion{Fe}{ii}      & 5284.109 &   $-$3.200 &FW06 &23317.63 &38.2 &  $-$3.63   & 40.6 & $-$3.86\\
\ion{Fe}{ii}      & 5291.661 &   $+$0.561 &K09&84527.76 &      &            & 42.8 & $-$3.95\\ 
\ion{Fe}{ii}      & 5306.182 &   $+$0.044 &FW06 &84870.87 &30.4  &  $-$3.70   & 26.7 & $-$3.91\\
\ion{Fe}{ii}      & 5315.083 &   $-$0.422 &K09&85048.61 &      &            & 13.2 & $-$3.93\\
\ion{Fe}{ii}      & 5316.615 &   $-$1.780 &FW06 &25428.78 &79.9 &  $-$3.71   & 81.9 & $-$3.93\\
\ion{Fe}{ii}      & 5318.055 &   $-$0.177 &K09&84527.76 &      &            & 22.3 & $-$3.85\\
\ion{Fe}{ii}      & 5355.421 &   $-$0.203 &K09&84685.20 &      &            & 13.2 & $-$4.17\\
\ion{Fe}{ii}      & 5358.872 &   $-$0.130 &K09&84685.20 &      &            & 14.8 & $-$4.17\\
\ion{Fe}{ii}      & 5359.237 &   $-$1.112 &K09&84710.70 &      &            &  7.2 & $-$3.59\\
\ion{Fe}{ii}      & 5366.210 &   $-$0.549 &K09&84710.70 &      &            & 16.4 & $-$3.68\\ 
\ion{Fe}{ii}      & 5425.257 &   $-$3.390 &FW06 &25805.33 &27.8  &  $-$3.59   & 30.2 & $-$3.81\\
\ion{Fe}{ii}      & 5430.003  &   $+$0.427&FW06 &85462.86 &       &            & 36.7 & $-$3.92\\
\ion{Fe}{ii}      & 5444.386  &  $-$0.153 &K09 &85495.32 &      &            & 19.5 & $-$3.90\\
\ion{Fe}{ii}      & 5451.316  &  $-$0.756 &K09 &84685.20 &      &            & 7.1  & $-$3.95\\ 
\ion{Fe}{ii}      & 5465.932  &  $+$0.348 &FW06 &85679.70 &42.3   &  $-$3.58   & 35.6  & $-$3.86\\
\ion{Fe}{ii}      & 5472.855  &  $-$0.723 &K09 &84685.20 &      &            & 10.5 & $-$3.77 \\
\ion{Fe}{ii}      & 5475.826  &  $-$0.129 &K09 &84685.20 &      &            & 19.7 & $-$3.96\\
\ion{Fe}{ii}      & 5488.776  &  $-$0.468 &K09 &85462.86 &      &            & 13.3 & $-$3.84\\
\ion{Fe}{ii}      & 5492.398  &  $-$0.127 &K09 &84685.20 &      &            & 25.1 & $-$3.78\\
\ion{Fe}{ii}      &  5493.830   &  $+$0.259 &FW06 &84685.20 &37.7   &  $-$3.68 & 32.5 & $-$3.93\\
\ion{Fe}{ii}      & 5502.670  &  $-$0.179 &K09 &85184.73 &      &            & 21.2 & $-$3.83\\
\ion{Fe}{ii}      & 5534.847 & $-$2.860   &FW06 &26170.18&50.5 &  $-$3.44   & 51.1 & $-$3.70\\
\ion{Co}{ii}      & 3501.708 & $-$1.111 & K06 &17771.506 &not obs& $-$ & not obs& $\le$$-$8.02\\
\ion{Co}{ii}      & 4160.657 & $-$1.751 & K06 &27484.371 &blend&$-$ & blend & $-$\\
\ion{Ni}{ii}      & 4067.031 & $-$1.834 & K03Ni &32499.530 & 9.9 & $-$6.24 & 9.8 & $-$6.47\\
\ion{Cu}{ii}      & 4909.734 & $+$0.790 & K03Cu  &115568.985& not obs & $-$& 3.1 & $-$6.26\\
\ion{Cu}{ii}      & 4931.698 & $+$0.704 & K03Cu  &115662.550& not obs & $-$& 2.7 & $-$6.22\\
\ion{Zn}{i}       & 4810.528 & $-$0.137 &K, WAR & 32890.352 &not obs & $-$& 5.1 & $-$5.85\\
\ion{Zn}{ii}      & 4911.625 & $+$0.540 &NIST3 &96909.740  &not obs & $-$&23.0 & $-$5.76\\
\hline
\ion{Ga}{ii}      & 4255.722 & $+$0.634 &RS94 & 113842.190&not obs & $-$ & not obs & $-$\\
\ion{Ga}{ii}      & 6334.069 & $+$1.000 &RS94 & 102944.550&not obs & $-$ & not obs & $-$\\
\ion{As}{ii}      & 5105.58  & $-$ & $-$    & 81508.925 &not obs & $-$ & 4.0     & $-$\\
\ion{As}{ii}      & 5107.55  & $-$ & $-$    & 82819.214 &not obs & $-$ & blend  & $-$\\
\ion{As}{ii}      & 5231.38  & $-$ & $-$    & 79128.330 &not obs & $-$ &  3.2 & $-$ \\
\ion{As}{ii}      & 5331.23  & $-$ & $-$    & 81508.925 &not obs & $-$ &  9.4 & $-$\\
\ion{As}{ii}      & 5497.73  & $-$ & $-$    & 78730.893 &not obs & $-$ & blend & $-$\\
\ion{As}{ii}      & 5558.09  & $-$ & $-$    & 79128.330 &not obs & $-$ &  9.4 & $-$\\
\ion{As}{ii}      & 5651.32  & $-$ & $-$    & 81508.925 &not obs & $-$ &  11.0 & $-$\\
\ion{As}{ii}      & 6110.07  & $-$ & $-$    & 83819.214 &not obs & $-$ & 3.3 & $-$\\
\ion{As}{ii}      & 6170.27  & $-$ & $-$    & 79128.330 &not obs & $-$ & blend & $-$\\
\ion{Sr}{ii}      & 4077.709 & $+$0.151 &NIST3 &     0.000&not obs & $-$ & not obs & $-$\\
\ion{Y}{ii}       & 3950.349 & $-$0.490 &K,HL &840.213&profile&$-$8.60& 4.2 & $-$8.20\\
\ion{Y}{ii}       & 4883.682 & $+$0.070 &K,HL &8743.316&profile&$-$8.60 & 7.3 & $-$8.03 \\
\ion{Y}{ii}       & 4900.120 & $-$0.090 &K,HL &8328.041&profile&$-$8.60 & 5.5 & $-$8.04\\
\ion{Xe}{ii}       & 4844.330 & $+$0.491 &NIST3 &93068.440&28.8&$-$5.23 & 17.5 & $-$5.81\\
\ion{Hg}{ii}      & 3983.890 & $-$1.510 &NIST3 &35514.000&profile&$-$8.20 & profile & $-$7.10\\
\hline
\end{tabular}
\par
\smallskip
$^{a}$ NIST3: NIST Atomic Spectra Database, version 3 at \url{http://physics.nist.gov};\\
FW06: Fuhr \& Wiese (\cite{Fuhr06});\\
PTP: Pickering et al. (\cite{Pickering02});\\
RS94: Ryabchikova \& Smirnov (\cite{Ryabchikova94});\\ 
SL: Sigut \& Landstreet (\cite{Sigut90});\\
K03Mn: \url{http://kurucz.harvard.edu/atoms/2501/gf2501.pos};\\
K03Ni: \url{http://kurucz.harvard.edu/atoms/2801/gf2801.pos};\\
K03Cu: \url{http://kurucz.harvard.edu/atoms/2901/gf2901.pos};\\
K06: \url{http://kurucz.harvard.edu/atoms/2701/gf2701.pos};\\
K07: \url{http://kurucz.harvard.edu/atoms/2601/gf2601.pos}, 2007 version, no longer available.
The $\log~gf$ gives better agreement with the observations than the K09 new determination;\\
K09: \url{http://kurucz.harvard.edu/atoms/2601/gf2601.pos}, January 2009 version;\\
``K'' before another $\log~gf$ source means that the $\log~gf$ is from Kurucz files
available at \url{http://kurucz.harvard.edu/linelists/gf100/}; 
(HL) Hannaford et al. (\cite{Hannaford82}); (KP) Kurucz \& Peytremann (\cite{Kurucz75}); (MRB) Miller et~al. (\cite{Miller71});
(NBS) Younger et~al. (1979); (WAR) Warner (\cite{Warner68}).\\
$^b$ The abundances given in parenthesis were derived from line profiles. All computed line profiles include
hyperfine components, except for that at 3917.318~$\AA$.    
\par
\end{table}